Author's response to reviews

Title: Relationship between Bone Mineral Density and Remaining Teeth and Its Related Physiological Factors in Postmenopausal Women

Authors:

Chang-Suk Kim (molar5103@hanmail.net)
Eun-Kyong Kim (jinha01@naver.com)
Kyeong-Soo Lee (drkslee@ynu.ac.kr)
Hee-Kyung Lee (lk3731@yu.ac.kr)
Youn-Hee Choi (cyh1001@knu.ac.kr)
Tae-Yoon Hwang (luke@ynu.ac.kr)
Jun Sung Moon (mjs@yu.ac.kr)

Version: 2
Date: 2 January 2015

Author's response to reviews:

Thanks a lot for advising our manuscript. 
first, we added e-mail address of all authors in title page
second, we added number of each line and page
third, we added list of abbreviations
If there are something to prove our manuscript, please inform us.
Thanks a lot again